Author’s response to reviews

Title: Growth, immune and viral responses in HIV infected African children receiving highly active antiretroviral therapy: a prospective cohort study

Authors:

Philippa M Musoke (pmusoke@mujhu.org)
Peter Mudiope (pmudiope@mujhu.org)
Linda Barlow-Mosha (lbarlow@mujhu.org)
Patrick Ajuna (pajuna@mujhu.org)
Danstan Bagenda (dbagenda@mujhu.org)
Michael Mubiru (mmubiru@mujhu.org)
Thorkild Tyelleskar (Thorkild.Tyelleskar@cih.uib.no)
Mary Glenn Fowler (mgfowler@mujhu.org)

Version: 8 Date: 7 April 2010

Author’s response to reviews:

In response to the editors request, I have divided figure 2 which has 8 graphs into 2 sections so as to allow uploading with manuscript